



Haoyu Wang <sup>1</sup>, Jianshe Ma <sup>1,\*</sup>, Yide Yang <sup>1</sup>, Mali Gong <sup>2</sup> and Qinheng Wang <sup>3</sup>

- <sup>1</sup> Tsinghua Shenzhen International Graduate School, Tsinghua University, Shenzhen 518055, China; haoyu-wa21@mails.tsinghua.edu.cn (H.W.); yyd22@mails.tsinghua.edu.cn (Y.Y.)
- <sup>2</sup> Department of Precision Instrument, Tsinghua University, Beijing 100084, China; gongml@mail.tsinghua.edu.cn
- <sup>3</sup> Shanghai Kulan Electronic Technology Co., Shanghai 200040, China

\* Correspondence: ma.jianshe@mail.sz.tsinghua.edu.cn

Abstract: The system-in-package (SiP) has gained much interest in the current rapid development of integrated circuits (ICs) due to its advantages of integration, shrinking, and high density. This review examined the SiP as its focus, provides a list of the most-recent SiP innovations based on market needs, and discusses how the SiP is used in various fields. Reliability issues must be resolved if the SiP is to operate normally. Three factors—thermal management, mechanical stress and electrical properties—can be paired with specific examples in order to detect and improve package reliability. This review provides a thorough overview of SiP technology, serves as a guide and foundation for the SiP in package reliability design, and addresses the challenges and potential for further development of this kind of package.

Keywords: integrated circuits; system-in-package; package reliability; advanced packaging; optimization

# 1. Introduction

Moore's Law, known as the "gold standard" of the chip industry, appears to have reached a "bottleneck" stage as a result of the ongoing advancements in integrated circuit technology [1–4]. Going Beyond Moore's Law has been suggested in this context [5–7], and the industry has given system-in-package (SiP) technology much attention. SiP technology combines numerous active devices that are based on bare chips with various passive devices that are all combined into a single package. A system-level device capable of performing specific operations is ultimately created through the processing procedure [8]. It is clear from the definition that the SiP is situated at the packaging's integration level. As indicated in Figure 1, there are three stages of electronic integration technologies now available: chip-level, package-level, and board-level [9]. The first level of integration is known as the system-on-chip (SoC), which is a deep integration of the system's components onto a single chip. The second level of integration is known as the SiP, which is a side-by-side or stacking of many chips [10]. To accomplish particular goals, the packaging is performed in 2D, 2.5D, or 3D. The board level, also referred to as the printed circuit board (PCB), is the third level.

Table 1 [11,12] displays a comparison of the traits of the three levels. The SiP design is based on the SoC [13], and the PCB is also required to execute the functionality when the SiP design is finished. It should be noted that the three levels are complementary to one another. As a result, it is impossible to fairly compare the three's performance in one particular area.

In fact, on the branch of packaging, besides the SiP, there is also the Chiplet [14]. The Chiplet, which is a small chip/core, is made by separating the components originally integrated in the same system monolithic wafer into multiple Chiplets with specific functions and then interconnecting them through advanced packaging technology to finally integrate



Citation: Wang, H.; Ma, J.; Yang, Y.; Gong, M.; Wang, Q. A Review of System-in-Package Technologies: Application and Reliability of Advanced Packaging. *Micromachines* 2023, *14*, 1149. https://doi.org/ 10.3390/mi14061149

Academic Editor: Xiaowei Wang

Received: 24 March 2023 Revised: 25 May 2023 Accepted: 26 May 2023 Published: 29 May 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).



the package into a system chipset. Chiplets are closely related to many advanced packaging technologies, so some people even consider Chiplets as an extension and expansion of the SiP.

Figure 1. Level classification of electronic integration.

1 /1	Table 1.	Comparison	of the three	levels in	electronic	integration.
------	----------	------------	--------------	-----------	------------	--------------

Characteristic	SoC	SiP	РСВ
Main material	Semiconductor	Conductor insulation	Conductor insulation
Volume	Small	Medium	Large
Cost	High	Low	Low
Period	Long	Short	Short
Туре	One chip	Multiple chips	Multiple chips
Function	Simplification	Diversification	Diversification

In heterogeneous integration, SiP technology is particularly important to achieve the goal of "Beyond Moore's Law". The SiP has value in almost every market segment. According to the Yole Group [15], SiP's share of hours exceeded USD 14B in 2020 and is expected to increase to at least USD 19B by 2026.

As shown in Figure 2, the SiP has evolved through three stages, namely original device packaging, SiP conventional packaging, and SiP advanced packaging. Original device packaging has DIP, LCC, SOT, and other inline or surface mount types of devices; SiP conventional packaging is the primary SiP packaging represented by flip chip sand wire bonding; in the new century, there are various more advanced packaging forms. It can be seen that the SiP is developing in the direction of systemization, functionalization, and diversification.

The SiP provides quick development cycles, high levels of flexibility, and low costs as compared to other packaging types, especially when it comes to optimizing the benefits of the system. The current increasing demand for high-performance packages has led to a preference for SiP technology [10], where a "combined" system may contain the functionality of many chips. The SiP is advancing in terms of density, power, and miniaturization along with the advancement of integration and packaging technologies [8,13]. SiPs include a variety of advanced packaging technologies such as the fan-out wafer-level package (FOWLP), chip-on-wafer-on-substrate (CoWoS), and others [14,16]. This review looked into some of the more common packaging techniques being used and identified several industrial domains where SiP technology is being used. The development of SiP technology has been accompanied by many reliability issues. To improve the overall performance of the SiP, thermal management, mechanical stress, and electrical properties need to be addressed [17]. This review looked into SiP design optimization techniques, as well as reliability fault detection schemes.



Figure 2. Trends in SiP.

The goal of this review was to thoroughly examine the creation, cutting-edge technologies, applications, reliability analysis, and optimization of SiPs. It is broken down into a total of six sections: Section 2 introduces the SiP's advanced packaging technology; Section 3 describes how it is used in various industrial fields; Section 4 provides the analysis and detection techniques for the SiP reliability issues; Section 5 discusses the idea of improving package reliability by optimizing the SiP design; Section 6 is a summary and outlook of the whole review.

#### 2. Advanced Packaging Technologies in SiP

#### 2.1. Interconnect

SiPs involve many interconnection cases, the main paths being wire bonding, solder joints and bumps, etc. Wire bonding is a method of connecting metal leads to the pads, which is equivalent to connecting the internal and external chips. From a structural point of view, the metal leads act as a bridge between the pads of the chip and the carrier pads. The former is the first bonding, and the latter is the second bonding. Solder joints are the conventional connection method and are more frequently used at the PCB level. Solder joints are usually used to connect the PCB or substrate with other electronic components. Bumps can be understood as reduced solder joints, often called a solder ball, which can be used to connect tiny devices.

The basic components of a SiP's design schematic are a combined chip, a silicon intermediary layer [18], a substrate, and a PCB, as depicted in Figure 3. The chips are integrated in a 2.5D or 3D shape as a stack. The two versions differ in that wiring and punching holes are performed directly on the chip in the latter, while wiring and punching holes are performed on an intermediary layer, which is inserted between the chip and the substrate in the former, such as through silicon vias (TSVs).

Wire bonding and flip chips are the two more popular varieties in the standard SiP procedure. As seen in Figure 4, the die/chip is wire-bonded to the substrate by these two methods. Wire bonding is utilized to attach the die to the substrate's top, and the flip chip is used to complete the interconnection by flipping down a conductive underfill, which has been placed right on the surface of the die to link it to the substrate. The flip chip is somewhat of an advancement over wire bonding, which primarily manifests in the following area: (1) Signal integrity (SI): The advantages of flip chips are more visible because the signal line path is significantly shorter than that of the wire bond, especially in high-speed signal scenarios with strict requirements for signal integrity [19]. (2) Power integrity (PI): The flip chip is smaller; the overall impedance from the power supply to the chip will be lower, reducing the voltage drop and power supply noise. (3) Flexibility: Flip chips have clear advantages in high-density connectivity situations, as they allow for connections throughout

Chip 1
Chip 2

Silicon interposer

Silicon substrate

the entire die's surface. (4) Miniaturization: Without additional room for connections, the high-density feature of the flip chip directly benefits system area reduction.

Figure 3. Typical SiP structure schematic.



Figure 4. SiP interconnection form: wire bond and flip chip.

Sophistication and diversification have also been present in the development of SiP technology. There are three further common forms of SiPs, namely flip chip, fan-out, and embedded die, according to the Yole Group's [15] analysis of system-level packaging technology and market trends. The fan-out and embedded die have clear competitive benefits over the flip chip packaging method in terms of structural volume, quantity, and the redistribution layer (RDL). They also perform well in terms of signal transmission and thermal management.

People have increased their demands for the manufacturing efficiency of the IC sector as a result of cost increases and technological advancements [20]. In order to increase the production efficiency of the entire sector, major corporations are gradually shifting toward a vertical division of the labor model in which design, manufacturing, packaging, and testing are separated. More importantly, the worldwide supply chain for SiP technology is extremely intricate and sophisticated, encompassing integrated design and manufacture (IDM), outsourced semiconductor assembly and testing (OSAT), and foundries. As a result, an increasing number of businesses are investing in research into integrated packaging, which keeps the SiP consistently inventive. Most of the OSAT companies, including ASE, Amkor, and JCET, which have achieved the top spot in packaging and testing, as well as the world's top IDM companies, Samsung Electronics, Intel, and Infineon, work to improve product competitiveness through SiP technology, particularly in the fields of the fan-out and embedded die. Universities and research organizations have also made contributions to the theoretical aspects of cutting-edge SiP technologies.

It has been discovered through extensive literature and website study that SiP advanced technologies are diverse.

### 2.2. Fan-Out Wafer-Level Package

The wafer is often chopped and sliced before being packaged according to customary packaging methods. Wafer-level packaging (WLP) involves placing an entire package on a wafer and then slicing it once it is finished. With this method, the package size can be close to the same as the bare chip size, which is consistent with the idea of electronic items being made smaller and lighter. There are two types of WLP: one is the fan-out type, which may be rewired outside the area where the chip is located to gain more pins, while the other one is the fan-in type, where the package's two-dimensional size is roughly equal to the chip's own size. The fan-out type becomes a better option as IC output pin counts rise.

The overall system density for FOWLP containers is very high. In order to modify the wiring technique corresponding to the original chip pins, which can be applied to more varieties of package forms, it is typically used in conjunction with the RDL [21,22]. Figure 5 depicts an FOWLP package based on RDL technology [23]. It consists of eight stages, including deposition, RDL fabrication, assembly, etching, and cleaning. The structure is designed to be extremely flexible, both as an overall SiP connected to the board level via solder joints and cut into individual chips for use. Theoretically, the RDL metal rewiring and bumping method can match the fan-out type and achieve the ideal layout. FOWLP thus addresses the issue of inadequate implant balls caused by too many chip pins in conventional wafer-level packages.





#### 2.3. Embedded Wafer-Level Ball Grid Array

The embedded wafer-level ball grid array (eWLB) is derived from the FOWLP [24]. Infineon created the first version of the eWLB by laying the qualified wafer face down on the larger wafer. The two are placed together into an epoxy mold before being finished with packing, wiring, dicing, etc. Many companies or organizations have improved and optimized their antenna packages after learning from Infineon's design architecture [25,26]. In order to use the eWLB, which is a fifth-generation packaging technique, the chip must

first be rearranged onto an artificial wafer, followed by fanning out to the chip's quad-axis, ball planting, and finally, packaging. This type of packaging not only boosts route density, but also reduces the package size, freeing up much board space. More crucially, the shorter interconnects and lower dielectric constant materials also result in better electrical performance. The eWLB package type offered by JCET has highly integrated 2.5D and 3D eWLB solutions [27,28]. The development process is quite thorough and includes conformal shielding, system-level testing, high-density surface mount technology (SMT), and advanced packaging processes. High-density interconnects in low-loss package architectures are made possible by eWLB-based intermediary layers created by JCET, resulting in more effective heat dissipation and processing speed.

#### 2.4. Integrated Fan-Out

Taiwan semiconductor manufacturing company (TSMC) created the integrated fan-out (INFO) for the first time in 2017. The FOWLP process is the foundation of the INFO, which completes the integration of many chips. With its compact dimensions and few connections, this package type prioritizes cost effectiveness. The INFO is appropriate for applications including the packing of high-performance computers (HPCs), high-frequency devices, and wireless chips. This form can provide space for numerous chips to be integrated at once [29]. A periodic electromagnetic band gap (EBG) structure in the form of an INFO package [30] is depicted in Figure 6. A cross-sectional view of this system is shown, along with micro-convex layer blocks, three different types of copper layers, and aluminum layers, and ultimately, the system is stacked on an SoC chip from top to bottom, with the metal layers connected by vias. The EBG is made up of a parallel-plate capacitor and two spiral-symmetric coupled inductors. Between Copper Layer 2 and Copper Layer 3, there are huge vias that connect the LC resonators. The system may deliver an excellent isolation environment with isolation levels up to 30 dB over a large frequency range (29.5 GHz).



Figure 6. Schematic of EBG in INFO package.

# 2.5. Double-Sided Molded Ball Grid Array

The majority of double-sided molded ball grid array (DSMBGA) usage is in 5G networks. The integration of fundamental analog, digital, and radio frequency (RF) functions is necessary for 5G network packaging. In order to reduce conduction loss with extremely complex processing and high-quality processing materials, it is essential to merge multilayer board production with fine wiring. The co-simulation of the electromagnetic interference (EMI), components, and packaging must be completed at the same time. Amkor originally presented the DSMBGA packaging concept [31] and realized mass production of the product, as illustrated in Figure 7 for a schematic of a DSMBGA package, in order to achieve this high-quality 5G integration. The strip milling, molded underfill (MUF) method is mostly used for the top and bottom filling. This method divides the shielding region to reduce electromagnetic interference and places additional power amplification circuits and filters to improve signal integrity.



Figure 7. Schematic of DSMBGA package concept.

#### 2.6. Embedded Multi-Chip Interconnect Bridge

Intel has made significant R&D investments in the embedded multi-chip interconnect bridge (EMIB) package. It is an organic-board-type package, unlike the other package forms, and lacks TSVs, which is based on a two-dimensional planar extension packaging technique. The whole EMIB package method is straightforward and high-yield since the EMIB is directly interconnected by silicon wafers for local high density [32]. This package offers good electrical qualities, a comprehensive shielding structure, and low crosstalk and loss. A very thin silicon bridge is placed in the top layer of the package system and connected to the chip pads of the substrate by vias in Figure 8, which is a schematic representation of the architecture of an EMIB [33]. The I/O's SI and power PI of the entire system are essentially unaffected by the EMIB's low and sparse pin count. When compared to typical package types, where the device must travel completely through the silicon intermediate layer, the EMIB's electrical properties are much better.



Figure 8. Schematic of EMIB system architecture.

Taken together, each of the above centralized SiP advanced packaging forms has its own characteristics, and the following Table 2 shows their comparison results [9].

The SiP forms are varied and show a convergence of numerous package advances so that they can also combine various advantages, whether it be the traditional wire bonding and flip chip or today's fan-out and embedded die. To summarize, the main goals of such sophisticated packages are to increase power density, reduce size and line length, enhance system performance, and lower system loss. Additionally, SiP technology is now based more on the requirements of the application scenario. Computers will take into account the system's processing speed; communication fields will take into account the signal's stability during transmission; power modules will take into account the system's ability to dissipate heat, etc.

Forms	Structure	Applications	Characteristics
			Low cost
FOWLP	2D	5G, AI [22–24]	Low parasitic effects
			High scalability
oW/I B	20/250/20	5C AI [25 28]	Embedded Easy wiring
evvLD	20/2.00/00	JG, AI [20-20]	Thin profile
	2D	HPC, smartphones [24,29]	Multi-chip integration
INFO			Systematization
			Multifunctional
			Unique package structure
DSMBGA	3D	RF, 5G [31]	Low power consumption
			Signal integrity
EMIB	2.5D	Graphics, HPC [32,33]	Good physical properties
			Easy processing
			Highly integrated

Table 2. Comparison of SiP forms.

#### 3. Application of SiP Technology

The fast growth of technologies such as artificial intelligence and the Internet of Things has raised the demands on chips and systems since electronic information has now permeated every aspect of our life. The incomparable benefits such as the SiP's tiny size, low power consumption, quick development time, and high adaptability are also present. As a result, the SiP is useful in an increasing number of fields. The application scenarios of SiP technology, such as wireless communication, power supply, central processing units/processors (CPUs), lasers, micro-electro-mechanical system (MEMS) sensors, etc., are outlined in this section.

### 3.1. Wireless Communication

SiP technology is now one of the key technologies in the field of wireless communication, where it was first used. The conventional packaging technology, which cannot satisfy the demands of high-performance devices, limits the ability of standard wireless systems to deal with the issues. By enabling the miniaturization and cost reduction of otherwise expensive and cumbersome devices at the system level, as well as in terms of dependability, SiP technology has emerged as a solution to these issues [34,35].

Millimeter wave refers to electromagnetic signal frequencies between 30 and 300 GHz, which broadly speaking comprises a variety of communication techniques such as radar, antennas, and sensors. The SiP of multi-layer/3D multi-chip modules is a more popular packaging technique [36], as it can combine the benefits of both thin-film and thick-film techniques with increased processing power and power quality [37]. With its adjustable size and shape in accordance with the actual application and perfect adaptability to the requirements of various components, as well as wiring, this upgraded method has low processing costs, high reliability, and high efficiency. As shown in Figure 9a,b for the five-layer and two-layer glass stacked structures, respectively, glass material has incomparable advantages for multilayer packaging of millimeter-wave antennas due to its good electrical properties and ability to be processed with high precision when compared to other materials [38]. The electrical connection between the glass and the peel is made using Cu and Sn. Layer-to-layer connections are made via a number of processes, including chemical gold plating, metal bonding, filling with plating, rewiring, seal ring sealing, and through glass vias (TGVs). With through holes for thick layers and blind holes for thin layers, the selection of holes is determined by the thickness of the layer. During testing, it was discovered that the antenna formed of a glass wafer has a more improved gain and is more stable when dealing with complex and varied antenna designs.



Figure 9. Glass lamination structure: (a) Five-layer lamination. (b) Double-layer lamination.

A ceramic substrate is another excellent option in addition to a glass substrate. The first one is a high-temperature co-fired ceramic (HTCC), which buries the vertically inflated linked waveguide and muffles the mixer's harmonic output by making use of strong mechanical strength and stable chemical strength. Radar arrays can use this technology because of its extraordinarily high transmission efficiency and guarantee of function realization [39]. The next application is a low-temperature co-fired ceramic (LTCC), which is for high-frequency communication. It was first proposed for semiconductor transceivers above 100 GHz, taking the mechanical defects of the LTCC into consideration [40]. The viability of the scheme was demonstrated after the completion of the robust design and standard assembly. The benefits of this material substrate for the SiP will be more obvious if the "hardware" weakness of the LTCC can be fixed.

#### 3.2. Power Modules

The majority of power electronics use power modules to supply power. Systemlevel packaging can resolve the seeming contradiction between the high power of power modules and the downsizing of size and weight [41]. The following difficulties should be taken into account when using the current SiP technology for power modules [42,43]: (1) realizing effective thermal control to enable efficient heat dissipation; (2) minimizing the parasitic effects brought on by the connection of various devices and enhancing the electrical characteristics; (3) possessing a certain capacity for external loads; (4) balancing the effects of the size, weight, and power loss. Since frequent vibration work environments are common for high-power modules in the automotive, aerospace, and other industries, the design of the package shape primarily considers mechanical stability. The assembled threedimensional structure of a power module, including a driver chip, MOSFETs, inductors, and discrete resistors and capacitors, is schematically depicted in Figure 10 [44]. There are two layers: the upper layer includes two MOSFETs, as well as extra resistors and capacitors, while the lower layer has four huge inductors in addition to the upper board's same components. Through a connecting board containing solder balls, the electrical connection between the two layers is made. The shapes and spacing of the solder balls can be modified to minimize the maximum stress and increase the dependability of the system through the use of finite-element analysis and actual solder ball measurements.



Figure 10. SiP structure schematic of power module.

# 3.3. Computers

Computers are innovating towards low power consumption, low cost, and high computational throughput. The idea of developing advanced computer processors and memories has shifted to SiP technology with smaller heterogeneous systems [45]. For instance, the European Microsystem Package 32-Bit Computer Module Unit [46] has a total mass of only 150 g and a power consumption of just 12 W while integrating a central processor, static data memory, dynamic data memory, and bus and clock resources. SiP technology has also been applied by a number of researchers to computer processor packaging. As shown in Figure 11, a silicon adapter plate is mounted on the package substrate using the CoWoS technology [47], which mounts multiple chips onto a silicon intermediate layer and interconnects them with high-density wiring. The CoWoS intermediate layer is attached to a single substrate to complete the package, which consists of two chips in total. The two chips are connected to the intermediate layer, where they are positioned below via bumps. The number of bumps in the intermediate layer is twice as many as those in the package substrate, giving the chips more places to be wired. The benefits of CoWoS are highlighted because the two chips are connected directly by a low-voltage-in-packageinterconnect (LIPINCON), and the data processing speed of this package can be increased to 8 Gb/s.



Figure 11. Schematic of the processor's CoWoS package.

#### 3.4. MEMS Sensors

The three pillars of information technology are MEMS sensors, communication technology, and computer technology. Conventional sensors are heavy and bulky, so they have high energy consumption. MEMS sensors are increasingly becoming more common since portable and miniature sensor devices are expected. SiP technology is one of the greatest options for MEMS sensor innovation because it is crucial for shrinking and lightweighting [48,49]. Figure 12 depicts the schematic architecture of a gas sensor [8], with the bottom substrate attached to the sensor by lead bonding and the top substrate fit with tube cores for temperature and humidity. A TSV loaded with conductive silver adhesive is used to electrically connect the substrate's bottom and top layers, while thermally conductive silicone grease is used to affix the remaining layers. The complete package system is far smaller than conventional printed circuit boards, making it more useful for evaluating specific situations with stricter volume restrictions. In addition to gas sensors, devices such as temperature sensors and ultrasonic rangefinders [50,51] can reduce the complexity of measurement while also increasing measurement accuracy by converting heavy equipment into small sensors.



Figure 12. Cross-sectional diagram of the gas sensor package.

# 3.5. Lasers

We always ask lasers to operate at higher luminous powers, wider arrays, and faster response times while still fitting them into small spaces [52]. The breakthrough in SiP technology for laser packaging is more necessary than ever. A small laser or other optical components are linked to the chip to create a complete system that is then connected to an electrical pin and soldered to a PCB [53–55]. This kind is different from the packaging for the numerous application cases outlined in the preceding section [56]. Figure 13 depicts an integrated package form of a femtosecond laser [57], with a silicon photonic chip on the left connected to 12 multicore fibers through 84 connecting channels in the center, while glass serves as the intermediary layer. The proximity fiber encapsulation is also schematically depicted in Figure 12, and it is created by laser irradiation and chemical etching to precisely match the fiber's hole diameter. The package also switches from a challenging five-axis connection to a simple single-axis connection thanks to the high accuracy of both methods. With its ability to properly balance loss, channel density, size, crosstalk, and other parameters, as well as its ability to reduce the burden associated with device assembly, these design and processing serve as a model for various coupling packages of micro lasers and precision photonic devices.



Figure 13. Femtosecond-laser-integrated package form.

As can be seen, the SiP is applied to numerous facets of our lives, further demonstrating the importance of this packaging type. However, the implementation of new processes and structures also introduces a number of reliability issues. If not detected and solved in time, it will make the system work abnormally and then lead to equipment failure or even major accidents.

# 4. SiP Reliability Analysis and Testing

All designed package structures must be analyzed for reliability in order to have application value. The reliability of the SiP refers to the likelihood that a product will fail or fail inside a specific phase and is a crucial metric for determining if the packaged product can function steadily over the course of a given life cycle. If we look at the failure form in terms of the research of package dependability, possible issues include warpage, chip cracking, delamination, toughness fracture, plastic deformation, and many more. If we break them down further, there may be hundreds of them, which makes it difficult to describe and raises the cost and testing effort. The dependability issue, however, may be broken down into three categories if we look for the source: thermal management, mechanical stress, and electrical properties. This way will make it simple to study reliability issues in an understandable manner and uncover the root reasons quickly based on the packaging failure form. In fact, there are many potential failure modes in the system, such as stress fracture, high-temperature deformation, high-temperature deterioration, open circuit, short circuit, line impedance mismatch, electromagnetic interference, and many other categories. Reliability mainly revolves around the above potential failure modes. The analysis and testing of temperature, mechanical stress, and electrical system-level reliability issues are summarized in this section.

#### 4.1. Thermal Management

Since no electronic device can now operate at 100% conversion efficiency, some electrical energy must be transformed into heat. With the unique advantages of small size and high density, the SiP helps to achieve system integration. Due to the stacked arrangement of multiple chips, the heat density inside the SiP is higher than that of discrete devices, and the heat transfer paths are further restricted in a limited and narrow space. According to the heat transfer equation, the temperature rise problem caused by the SiP's special structure is more severe than that of packages such as TSOP.

$$Q = kA \frac{\Delta T}{\Delta x} \tag{1}$$

where Q, k, and A are heat, the heat transfer coefficient, and the heat transfer area, respectively.  $\Delta T$  and  $\Delta x$  are the changes of temperature and distance, respectively.

The heat dissipation of the entire system will be impacted and the internal temperature will rise if the thermal conductivity of the encapsulation system is not resolved. The mechanical stress will change irregularly as a result of excessive differences in the coefficients of thermal expansion (CTEs) of adjacent materials. In addition, as temperature rises due to ongoing heat buildup, component performance is also impacted, which in turn affects the system's overall electrical properties. As a result, the majority of reliability issues are caused by temperature issues, which necessitate thermal characterization analysis.

#### 4.1.1. Thermal Characterization Methods

Currently, thermal characterization can be accomplished in two steps: simulation and measurement. A few popular techniques are given below:

• Finite-element analysis (FEA):

In terms of thermal analysis, finite-element analysis has advanced considerably. There are other simulation programs besides ANSYS, Comsol, Flowtherm, etc., that can be utilized for the thermal characterization of the SiP, whether it is a 2D or 3D structure. According to the survey, the SiP applications of FEA are more likely to restore the functioning scenario of the SiP [58–61], such as by adding suitable temperature cycles, simulating the architecture made up of silicon vias and bumps, modeling reflow soldering, etc. Modeling is a must for finite-element analysis, but the 3D kind of SiP is complicated, and if the simulation's accuracy is to be ensured, the computational time will undoubtedly grow and the efficiency will be significantly decreased. The global multi-structure 3D model is reduced to a 2D structure with separate structural coupling connections [62] when a quasi-3D structure is utilized, which can meet the accuracy requirements and speed up FEA computations. In general, FEA methods still address modeling problems, especially concentrating on the cross-scale and cross-dimensional aspects of 3D models.

Thermal imaging technology:

Thermography is the act of turning thermal radiation into an electrical signal that can then be amplified and processed to create a thermal image that shows the temperature distribution throughout an object's surface. Thermography can be used to measure the local temperature of microstructures such as the SiP or MEMS [63], with the benefit of being quick and easy to use, but the drawback of only being able to assess surface temperatures.

Phase-locked infrared thermal localization:

Phase-locked infrared thermal localization overcomes the limitations of two-dimensional measurement and may find deep hot spots to address the inadequacies of thermal imaging methods [64]. This method is an excellent complement to thermal analysis for 3D designs and employs computer simulations to find flaws inside 3D stacked structures via surface imaging. The drawback is that the phase-locked infrared localization algorithm is more complicated and takes a while to compute.

• Thermal resistance matrix method:

If a single heat source calculation is still performed, SiP technology's multi-chip stacking will surely result in several heat sources throughout the entire system, which will result in a considerable difference between the simulation results and the actual ones. In order to quantify the coupling and mutual influence between various heat sources with high reliability in thermal characterization, the thermal resistance matrix of the 3D stacked package must be created [65]. In general, the accuracy of this method improves as more chip parameters (temperature, structure, material, and electro-thermal characteristics) are collected.

Scanning:

The scanning method is another adapted concept used to describe the SiP's thermal characterization. With high testing efficiency and accuracy, X-rays can be utilized to scan the entire packaging system and complete the structural and functional testing, which is extremely valuable for high-precision packing equipment.

# 4.1.2. Thermal Characterization Objects

Presently, researchers studying thermal characterization are mostly concentrating on substrates and ball grid array (BGA) solder connections, both of which are important structures that have an impact on the SiP's thermal dependability.

A guarantee for the appropriate operation of the entire system is if thermal issues can be properly resolved because the substrate serves as a crucial carrier for the electrical connectivity and physical support of the SiP. At this point, simulating the substrate is necessary both alone and in conjunction with connected components. Computational fluid dynamics (CFD), for instance, is utilized to examine the interconnection of the silicon chip and LTCC substrate and the evolution of the bonding process between them [66]. According to research on the thermal properties of three different package types—2.5D package, fan-out chip on substrate (FOCoS) chip-first, and FOCoS chip-last—the junction temperature of the FOCoS type is lower than that of the 2.5D package, meaning that the substrate material and the design of the structure both have effects on the thermal properties [67]. The two FOCoSs' shared grinding operation, which uniformly thins the material between the heat sink and the package and, so, ensures overall thermal efficiency, is the primary cause of this.

BGAs meet the requirements of the SiP because they feature a large number of pins, a high yield, and good electrical characteristics. Yet, in order to link devices to the PCB, BGAs are inescapably exposed to heat-related stressors, which might in turn cause reliability issues. The present research strategy for BGAs' thermal issues is straightforward: typically, the architectural model of the system and the parameter details of each device are first established, and then, simulations in finite-element software are performed to obtain the related thermal results. Following completion, measurements are taken using machinery to confirm them. The difference between the model utilized and the actual computed findings was less than 2 °C in the low-temperature range and less than 4.5 °C in the high-temperature range, according to control tests carried out by certain researchers using a thermal imager [68]. The accuracy of the finite-element solution has increased, and the modeling parameters have been continuously subdivided and supplemented. As a

consequence, the thermal simulation results of the BGA closely match the actual values, which paves the way for further BGA optimization.

#### 4.2. Mechanical Stress

Another important factor in determining the dependability of SiPs is the examination of mechanical stress. In system-level packages, the materials used for the chips, substrates, leads, and solder connections vary. Different materials will shrink or expand in accordance with their unique qualities when subjected to influences such as temperature change or external impact, leading to specific deformation and consequent mechanical stress issues. As an illustration, we give the formula for the CTE:

α

$$=\frac{\Delta l}{l \times \Delta T}$$
(2)

where  $\alpha$ , l, and  $\Delta$ l are the CTE, length, and length change, respectively. When two materials are heated, their CTEs differ, the strain will also be noticeably inconsistent, and a certain amount of stress will be produced, which falls under the category of mechanical stress. Loss failure and overstress failure are the two primary types of failure brought on by mechanical stress. The former refers to the buildup of low stress that is continuously applied over an extended period of time and results in component loss, leading to system failure; the latter refers to failure brought on by a great stress that exceeds what the components can withstand during a specific event or moment. More focus has been placed on loss failure because, as was previously mentioned, a significant portion of the mechanical stress is caused by an imbalance in the CTEs of various materials as a result of the temperature rise. Therefore, it is more advantageous to investigate the specific causes of failure. Moreover, loss failure is more valuable to examine because it occurs frequently in all packing systems, whereas overstress failure is more like an unforeseen, uncontrollable catastrophe.

#### 4.2.1. Stress Measurement Methods

Lossy testing and nondestructive testing are two different ways to evaluate stress. The latter is non-destructive, flexible, reliable, and dynamic, which is better suited for system-level packaging, which requires meticulous and exacting testing. The following non-destructive testing techniques are widely applied:

• Diffraction method:

The diffraction method uses the properties of X-rays, which diffract when traversing materials such as polycrystalline metal surfaces. In the event of stress, the resulting lattice distortion means the stress of the relevant material in accordance with Hooke's and Bragg's laws. This technique is well suited for use in 3D integrated measurements and can quantify both macroscopic and microscopic residual stresses [69].

• Indentation method:

Impact loading is used in the indentation method to make an indentation in the center of the strain flower, and resistive strain gauges serve as the sensitive element for measurement. A strain gauge records the change in strain increment in the indentation area and uses that information to calculate the stress value. To examine the TVS stress analysis of the SiP, the indentation method has recently been enhanced to the nanoscale level with extremely high stress resolution and displacement resolution [70].

Curvature method:

The level of constructive stress indicates the magnitude of strain at the material interface of each layer, while curvature is a quantitative measure that reflects the degree of curvature of a curve or surface. At the microscale, residual stresses normal to the material interface are calculated using the curvature method, and a thin film is chosen as the measurement object. If chosen properly, this measurement method is simple, effective,

and highly accurate [71]. However, if the chosen film cannot replace the original stress scene, the measurement method may have some flaws.

Blind hole method:

The blind hole technique entails making a tiny hole in the region that needs to be measured. The stress field here changes as the stress close to the hole is released, and by measuring the new value, the original stress value may be deduced [72]. Almost all types of materials in SiPs can be used with this technology, but the resolution of the stress measurement might not be as good as with the above methods.

#### 4.2.2. Mechanical Stress Analysis

Two ideas can be used to analyze the mechanical stress reliability of SiPs: one from the study's subject and the other from typical failure types.

The SiP structures that are susceptible to mechanical stress are solder joints, bumps, and package boards. Bumps are similar to solder joints, which play the role of connecting the substrate and device. Due to the mismatch of the thermal expansion coefficients of different materials, this will lead to both being susceptible to mechanical stress. After a period of accumulation, the continuous mechanical stress brought about by the temperature increase will reduce the working life of the device. Thermal cycling tests are typically performed to create stress or strain maps of the solder joint array for the tropicalization stress problem, which is then followed by finite-element analysis [73]. In general, the most-stressed solder joints in embedded SiPs are found in the middle of the package, whereas they are found towards the margins of the package in the standard SiP. In addition to thermal cycling, the impact of vibration on the solder joints or bumps has also been investigated using a type of simulated random vibration [41]. To determine the associated danger spots, a secondary simulation of the calibrated model is run with the model's parameters changed. To determine how long a solder joint will last, some of them can be incorporated into a life prediction model [74]. Traditional testing techniques, including temperature shock tests, temperature and humidity deviation tests, etc., are used to test the stability of package boards-typically SiPs with a laminated chip embedding. At high temperatures, specific types of stress manifestations, such as those brought on by local temperature gradients, can lead to higher expansion of the packaging system, which needs intermittent service life testing. Conventional stress testing techniques typically focus on the impact of a single stress, but in reality, failures are frequently caused by the coupling of pressures brought on by a number of different variables. Because of this, some people have adopted the strategy of combining stresses to create novel stress measurements, such as investigating the effects of both thermal cycling and humidity, i.e., the superposition of both forms of stress on the filler material [75], which frequently more accurately depicts the actual package stress situation.

Warpage, deformation, and other problems are the principal mechanical stress-related failures of the SiP. The amount of an arbitrary surface's out-of-plane displacement is measured using the term "warpage". It is essential to reduce the warpage phenomena of SiPs since wafer-level packaging technology is making the package material thinner and thinner [76]. Figure 14 displays a warpage demonstration for an epoxy molding compound (EMC), with two routes, AB and CD, which may calculate the warpage, as well as the maximum and minimum warpage axes [77]. Solving the CTE of an unknown structure is a necessary step in the analysis of the failure form. In order to determine the CTE before and after each process, researchers have adopted a new idea of measuring the CTE by using the thermal profile of a similar process to analyze the thermomechanics. This method is known as process thermomechanical analysis (pTMA). It also makes it possible to ascertain whether or not the changes brought on by temperature are reversible. As a result, pTMA seems to be a reliable analysis technique for CTE-related dynamic mechanical stress analysis. The primary cause of the deformation is the stress shock, which 3D SiPs still need to identify. The sensors that can measure stress and vibration in the folded region of the face to face are often present in the more widely utilized cubic SiPs in recent years. By incorporating the system into a package and simulating a typical working situation with shock or vibration while using the embedded sensors for evaluation, the reliability of the cube package has been confirmed [78].



Figure 14. EMC warp demonstrator.

# 4.3. Electrical Properties

The SiP's reliability concerns ultimately manifest as problems with the electrical properties as a result of the thermal and mechanical stress problems. Examining the electrical performance is another technique to determine the SiP's dependability or to see if it functions as intended. The electromagnetic environment of the entire package system is extremely complex because of the increasing density of SiPs, and the interconnection channels' effects on the signal are becoming more and more important. It is crucial to take into account a number of factors while evaluating electrical performance, including EMI, SI, and PI.

4.3.1. Electrical Properties Measurement Methods

- The following can be used to categorize the analysis and testing of electrical properties:
- Model construction method:

In order to understand and solve the electrical parameters, the model-building process typically includes replacing a complex packaged system with a suitable circuit model. The electromagnetic theory of Maxwell and fundamental circuit models are some of the modeling foundations. The majority of electrical-performance-related aspects can be addressed using this methodology. A double-sided component placement (DSCP) model based on RLC circuits can be created for PI [79]. In order to assess the performance of the entire power distribution network (PDN), the DSCP model includes the impedance analysis from the power supply to the motherboard and from the regulator to the IC, as illustrated in Figure 15. Some researchers have tested electromagnetic interference using a progressive RF optimization method [80]. A reference for the precise analysis and optimization of the near-field coupling between lines can be found in the use of dual-interconnected lines as a model to study the characteristics of near-field coupling and the establishment of an equivalent aggregate circuit model to specify the complex model with devices. In order to solve the circuit's connectivity security, a novel measuring concept has been put out [81].

The common method of measuring open-/short-circuit problems in the RDL is given a model, to which a simplified diagram of the measurement system, including insulation resistance, capacitance, converter gain, etc., is added. By analyzing the two DC signals given by the former level circuit, it is possible to determine whether there is an open or short circuit. The whole process can be performed by using mixing technology and a few passive devices, which greatly simplifies the testing steps.



Figure 15. DSCP model based on RLC circuit.

System integration method:

Using external devices for testing still presents some challenges because the SiP is a sophisticated system. Direct integration on the SiP has been suggested by researchers, including the joint test action group (JTAG) [82]. The JTAG can be tested directly through the SoC in the multiplexed system for the testing of SiP devices. The transient dose rate effect (TDRE) of the SiP [83], which results in the combination of electrons and holes, producing photocurrents, which subsequently interfere with the circuit, is treated similarly. The response method involves programming and recording the characteristics and locations of abnormal signals in the circuit using a field-programmable gate array (FPGA), which is already built into the SiP.

• Numerical calculation method:

A complex computational issue, the analysis of the electrical properties of SiPs calls for advances in numerical computation, increasing the effectiveness of data solving [84]. For instance, there are billions of unknown quantities in the high-performance JEMS-CDS software built on the JAUMIN platform that need to be resolved. The solution speed can be significantly increased by using the orthogonal matrix and region decomposition preprocessing approach. For issues such as EMI and SI, when the solution volume is enormous and the computing domain is divided, this kind of method is particularly useful.

Scanning method:

The scanning technique is also a more typical way to examine the package in question's electrical characteristics. For electromagnetic interference issues, near-field scanning can be performed to determine whether the SiP's shielding function is efficient [85]. The SiP substrate's dielectric constant has a significant effect on the signal in high-frequency scenarios in terms of SI, and two-dimensional scanning can be utilized to test the substrate's dielectric constant as well [86].

### 4.3.2. Electrical Properties Objects

The fourth-generation of the interconnect technique, the TSV, is the primary structure for achieving vertically oriented connections in SiPs [87]. To confirm the reliability of more TSV structures as they are created, related electrical performance testing tools are required. For instance, the RLCG parametric analytical formulation equivalent circuit model was developed to study the dependability of strip TSV structures, and simulations with ADS and HFSS were used to compare and validate the model's viability. The impact of the size of the strip TSV on the signal integrity was examined [88] in order to emphasize the benefits of the strip structure. As shown in Figure 16a, the longer the long side, the smaller the cross-sectional area, which can make the leakage current decrease and, thus, reduce the insertion loss, given that the sum of the circle diameter and the long side of the bar TSV is constant. As seen in Figure 16b, using a strip TSV with a long-edge feature can enhance the system's SI. Furthermore, specific cases have been investigated using millimeter-wave test structures to examine the transmission characteristics of a single TSV, a dual-redundant TSV, and a quad-redundant TSV in the presence of specific system faults [89]. The HFSS software has been used for simulations to show that the redundant class of TSVs can operate autonomously at high frequencies (44 GHz), even in the presence of fault issues. In the experimental portion, the RF redundant TSV test apparatus was constructed, and the insertion loss at 40 GHz for the dual-redundant and quad-redundant TSVs, respectively, was measured to be 0.19 dB and 0.46 dB, with 0.22 dB for the single TSV. The ability to handle faults is improved by the redundant TSVs, despite the fact that the insertion loss grows as the number of TSVs does.



**Figure 16.** Research of strip TSV structure: (**a**) Schematic diagram of strip TSV structure. (**b**) Comparison of insertion loss of strip TSV.

After mastering the reliability analysis of the SiP in three aspects: thermal management, mechanical stress, and electrical properties, we can try to propose an optimized solution based on the causes of failure.

#### 5. SiP Reliability Optimization Solution

The optimization objectives in this section can also be divided into three categories: the goal of thermal management is to limit the maximum temperature to less than 120 °C during a period of normal operation; the goal of mechanical stress management is to reduce the maximum mechanical stress value as much as possible; electrical performance will be further divided into a voltage drop less than 5% of the maximum supply voltage, a current density less than 30 A/mm<sup>2</sup>, an SI important signal return loss less than -40 dB, and EMI indicators required to be less than 1 v/m.

### 5.1. Thermal

Three basic methods have been proposed to address the SiP thermal issue: the addition of a heat sink; the use of materials with high thermal conductivity; the optimization of the SiP structure.

# 5.1.1. Adding Heat Sinks

The high-density system must contain a heat dissipation part, and the temperature rise caused by the operation of high-power devices is solved by heat sinks or other heat dissipation methods [90]. Monolithic microfluidic cooling, which is a low thermal resistance silicon fin engraved on a small chip, is an example of a cooling solution applied to small sizes [91]. Figure 17 depicts a schematic of monolithic microfluidic cooling, which places microneedle fin heat sinks on the chip and adds manifolds around them to facilitate coolant flow and transfer. This small heat sink, which has a thermal resistance of 0.07 °C/W and can maintain a constant temperature of 30 °C in an FPGA system with a power of 107 W, is more practical in space-dense systems than huge, bulky heat sinks. Some researchers have compared the shapes of the same type of radiators and found that tree-fin-type radiators are superior to parallel fins and interlaced fins [92]. Since the tree fins allow the fluid between adjacent channels to flow through one channel first and then separate into two channels, it is equivalent to an equalizing effect of heat. This method can reduce the temperature difference inside the system, prevent the problem of uneven heating of the device, and facilitate the overall heat dissipation.



Figure 17. Schematic diagram of monolithic microfluidic cooling.

#### 5.1.2. Using High Thermal Conductivity Materials

The solution to high temperatures is to conduct heat out quickly, so the use of highly thermally conductive materials is also a viable method. In order to function properly, pulse sequencers functioning in hot conditions urgently need to channel much heat away from them. To provide good thermal conductivity, an ultra-thin conductive adhesive is typically applied between the substrate and the SiP's base. Decoupling capacitors, which also have a high thermal conductivity, are one of the COG materials [93]. Inadequate thermal management can cause deformation and failure in solder joints [94]; to increase reliability, filler adhesive materials can be used, such as epoxy resins with additional components such as boron nitride and aluminum trioxide, which have strong temperature resistance and thermal conductivity. According to the experimental findings, the inclusion of filler glue increased the fatigue life of solder junctions by a factor of three [95].

### 5.1.3. Optimizing SiP Structure

To overcome the problem of thermal reliability, one can study the thermal reliability from the design structure of the SiP itself in addition to the typical addition of heat sink and thermal conductive materials. In the SiP system, there will be numerous integrated chips, which means numerous heat sources. The thermal reliability of SiPs can be increased through logical layout and optimization at the two-dimensional plane or height level [96]. The best outcome, using the optimization of dual-hot spots as an example, is to be able to logically arrange the two heat-generating chips so that the heat dispersion between each layer is very consistent. By building a heat transfer model of the 3D temperature field, a neural network is used to obtain the intrinsic connection between the hot spot map and the different chip placement positions. In order to find the ideal hot spot location, a multi-objective genetic algorithm can be used to solve the objective functions of the maximum temperature and the uniformity of the temperature distribution. Figure 18 illustrates the general symmetry of the temperature distribution, which strikes a reasonable balance between the maximum temperature and the even dispersion of heat [97]. Vias also improve the thermal performance, so adding grounded vias around the stacked chips is a good option. The heat is transferred to the metal layer through the vias and then dissipated using air cooling or convection [98]. The thermal dependability of the packing system can also be impacted by factors including the filling material, diameter, thickness, and spacing of the TVSs [99]. After simulation and experiment, it was discovered that, on the one hand, we can increase the thickness of the TVS high-thermal-conductivity filler material and decrease the TSV spacing; on the other hand, we should decrease the number of TSV chips used and increase the chip size, properly.



Figure 18. Temperature distribution diagram of optimized layout hotspots.

#### 5.2. Mechanical Stress

In the vast majority of cases, mechanical stress and thermal management work together to cause failures such as warpage, plastic deformation, and fatigue fracture. Mechanical stress between different materials is caused by the different CTEs due to the rise in temperature caused by the operation of the system. Therefore, the above-mentioned increase in thermal reliability partially resolves the issue of mechanical stress. The solder pins and lead bonding of SiP technology are also the more serious reliability difficulties, in addition to the thermal–mechanical reliability challenges.

Aluminum nitride has a high thermal conductivity and a similar CTE to the chip, making it an ideal material for making substrates. The reliability issues that need attention are mainly focused on the strength of the connection between the aluminum nitride substrate and the metal pins. Figure 19 depicts a model diagram of the established stress test, which can be used to investigate the variables influencing the solder strength of the pins [100]. The substrate is aluminum nitride, and the two lead types can be investigated for vertical tension and peel force, respectively. The vertical lead is a nail head lead soldered to a circular pad, and the horizontal lead is a flat lead soldered to a square pad. It was discovered through testing that increasing the thickness of the nickel plate layer and the quantity of brazing material used for soldering would result in a reduction in the strength of the pin soldering. The strength of the pins is also influenced by the brazing's structure. By comparison, the strength of the soldered pins can be significantly increased by using a soldering structure with bent pins.



Figure 19. Tensile test structure.

Copper-to-copper/silver bonding is one of the key technologies in the packaging section of SiP technology, which depends on the bonding technology to achieve chip interconnects [101]. The bonding strength can be significantly increased by creating a copper nitride coating on the copper surface using a two-step plasma process, which can produce low-temperature copper bonding and a shear strength of up to 30 MPa [102]. Figure 20 depicts a schematic diagram of the two-step plasma treatment [103]. The silicon wafer is first covered with a copper layer, and the deposition method's addition of a reinforced copper film with an argon gas stream is the first plasma treatment phase. When the surface impurities are cleaned up, the second plasma treatment phase is carried out, utilizing nitrogen gas to stop the copper coating from oxidizing in the presence of oxygen. This process improves the mechanical durability of the SiP by providing the bonded copper with outstanding passivation capability, stability, and high shear strength.



Figure 20. Two-step plasma treatment process.

# 5.3. Electrical Properties

Electrical performance has a fairly wide range, so dependability issues should be taken into account with the scenario's particular application. For instance, the SI is particularly crucial for high-frequency signal transmission; for power modules or DC–DC, the general focus is on PI; EMI is the main area of examination when numerous independent modules are combined to provide system operation.

### 5.3.1. Signal Integrity

SI issues can easily occur during signal transmission if the transmission line's impedance is discontinuous, which can easily lead to reflection phenomena. Transmission lines, vias, and solder joints on the substrate make up the whole signal transmission path in the SiP, and an evident impedance discontinuity is present. The SiP structure must be tuned to make impedance matching in order to increase SI, which includes choosing substrates with strong RF properties and a low dielectric constant as the adapter board, optimizing the size and arrangement of solder connections [104]. Impedance matching can also be achieved in the machining process, such as the use of shaped ramp interconnection techniques, which allow impedance matching in the K and W bands [105]. As an illustration, Figure 21 displays a schematic of a two-dimensional cross-section of a ramp interconnect that was inkjet-printed on a MEGTRON 6 substrate [106]. The tube core is put in place; the substrate is initially bonded with ink, and then, the attachment is finished with ultraviolet light. Immediately after, four layers of ink are printed, and ultraviolet light is still used to cure the deposition ramp that connects the substrate to the top of the die. The full ramp can be used for multi-layer inkjet printing and has a maximum slope of 35 degrees. The insertion loss of each ramp connector after this procedure was empirically confirmed to be 0.45 dB/mm, which is entirely acceptable in the high-frequency band.



Figure 21. Schematic diagram of the two-dimensional cross-section of the ramp interconnection.

#### 5.3.2. Power Integrity

The power supply network is made up of decoupling capacitors, the ground, and power. Parasitic effects are easily present in high-speed or high-frequency environments, causing voltage drop and power supply noise, which can further impair the chip's power supply or the transmission of other crucial signals. Increasing the width of the power supply alignment, rationalizing the device placement, minimizing the parasitic effects, and shortening the distance between the power and ground are typical ways to increase PI. A more representative case for improving PI in SiPs is in the package design of inductive regulators. This is usually performed by placing the regulator as close to the load as possible so that the power management loop is shorter and the parasitic effects are minimal, improving power integrity [107,108]. Inductive regulators can be packaged using an embedded planar manufacturing technique. A printed circuit board is filled with a composite magnetic material that is compatible with the board-level material and is etched in accordance with the inductor's size. The self-resonant frequency can be raised by 3.4% with this technique, and the quality factor can be raised by 20%. A TSV-based electromagnetic bandgap structure has been suggested by other researchers as a method of assisting the PDN in achieving broadband noise suppression. In comparison to the standard EBG, the EBG with the integrated TSV device may be employed in a gated PDN with a maximum rejection bandgap of up to 13.22 GHz. The bandgap can be altered by altering the structural parameters for various scenarios, which is stable and useful [109].

#### 5.3.3. Electromagnetic Interference

The EMI problem stems from the fact that the interference source passes through a coupling path and transfers energy to other sensitive components. There are three forms of implementation: conduction, radiation, and near-field coupling. It is usually overcome by the rational layout of the power supply and ground layer, partitioning of the same type of devices, and increasing the shielding structure. The more widely used research has concentrated on the topology and addressed the present EMI problem for the integrated package. For instance, when applied to the single-branch LLC resonant converter's high-voltage power supply architecture, the performance of the converter declined as the number of turns ratio increased. The secondary can also be linked to the relevant load if the configuration is altered to a scalable distributed LLC topology by cascading multiple primaries and secondaries in order to eliminate the voltage drop on each branch's primary winding. With this method, the turns ratio can be reduced to achieve a high buck ratio conversion by increasing the number. The distributed LLC's maximum radiative power is 24 dBm, and its maximum electric field strength is only 210 V/m, or 28% of a single branch, according to an EMI near-field simulation, which substantially resolves the EMI problem [110]. Additionally, it is worthwhile to investigate the package voltage regulator (VR) on the SiP placement, as shown in Figure 21. Both structures contain four components: an IC, a VR, a SiP system, and a decoupling capacitor at the bottom; the red and black lines denote the 12 V and 0.8 V power supply networks, respectively [111]. The only difference is that Figure 22a places the VR on the top level and Figure 22b places the VR on the bottom level. The former offers additional layout benefits because it can both minimize the direct connection distance to the IC and free up space at the bottom for the standard arrangement of BGA pins. The bottom configuration has relatively substantial EMI problems, according to simulation findings from ANSYS SIwave, because the VR is closer to the BGA, causing a larger direct power network from the linked power pins. This ideal layout method on the topology of independent devices can successfully address the system's EMI problems.



Figure 22. Topology of VR in SIP: (a) Top-level placement. (b) Bottom-level placement.

### 6. Conclusions

Future advancements in circuit technology will make them more streamlined and integrated, raising the bar for packaging technology. SiP technology is a good fit for these requirements in terms of definition and practical use. More importantly, the current competitive pressure in the packaging industry is such that companies are striving to be able to achieve high-volume and high-reliability production at a low cost. A known good die (KGD) for multi-chip assemblies is a good way to do this and, to a certain extent, also provides favorable conditions for SiP innovation. In this review, the history and

advancement of SiP technology were discussed in relation to the current broader context, and the features and benefits of SiP technology were compared and analyzed. After investigating the advanced SiP technology of a large number of well-known companies, we summarized the fields of the SiP's applications, including communication, lasers, MEMS, power supply, etc.

SiP technology advancements also result in issues with package dependability. This review outlined the techniques for assessing the SiP's reliability and, in addition, focused on the elements impacting reliability. The reliability difficulties were broken down into three categories: thermal management, mechanical stress, and electrical properties. Finally, it looked at how to increase package dependability from the three perspectives, discussed how to improve SiP designs, and provided practical examples that might serve as appropriate benchmarks for future SiP designs.

SiP research will keep concentrating on developing novel package structures and enhancing package reliability for some time to come. Both a challenging research question and the packaging industry's future development trend revolve around how to maximize performance under the constraints of volume and weight limitations. If reliability can be further ensured, the SiP will have a broader development prospect in various industrial fields.

**Author Contributions:** Conceptualization, H.W. and Y.Y.; methodology, H.W.; validation, J.M., M.G. and Q.W.; formal analysis, H.W.; investigation, H.W.; resources, H.W.; data curation, H.W.; writing—original draft preparation, H.W.; writing—review and editing, J.M.; project administration, J.M.; funding acquisition, J.M. All authors have read and agreed to the published version of the manuscript.

**Funding:** This work was supported by the National key research and development program "Information Photonics Technology" Key Project (Grant No. 2022YFB2804302). The funder is the State Key R&D Program Organization and is funded by the central government. The research of the funding number is about the drive and servo control.

Data Availability Statement: No new data were created.

Conflicts of Interest: The authors declare no conflict of interest.

### References

- 1. Borkar, S. Obeying Moore's Law beyond 0.18 Micron [Microprocessor Design]. In Proceedings of the 13th Annual IEEE International ASIC/SOC Conference (Cat. No.00TH8541), Arlington, VA, USA, 13–16 September 2002; pp. 26–31. [CrossRef]
- 2. MacK, C.A. Fifty years of Moore's law. *IEEE Trans. Semicond. Manuf.* 2011, 24, 202–207. [CrossRef]
- Jin, L.; Brocade Communications. T-5 What Makes Moore's Law Continue?-Recent Advances in Semiconductor. In Proceedings
  of the 2007 7th International Conference on ASIC, Guilin, China, 22–25 October 2007.
- 4. Mack, C.A. The end of the semiconductor industry as we know it. Emerg. Lithogr. Technol. VII 2003, 5037, 1. [CrossRef]
- 5. Zeng, S.; Tang, Z.; Liu, C.; Zhou, P. Electronics based on two-dimensional materials: Status and outlook. *Nano Res.* 2020, 14, 1752–1767. [CrossRef]
- Graef, M. Positioning More Than Moore Characterization Needs and Methods within the 2011 ITRS. In Proceedings of the Frontiers of Characterization and Metrology for Nanoelectronics, Grenoble, France, 23–26 May 2011.
- Pizzagalli, A. Lithography technology and trends for More than Moore devices Advanced Packaging & MEMS devices. In Proceedings of the 34th European Mask and Lithography Conference, Grenoble, France, 18–20 June 2018.
- 8. Santagata, F.; Sun, J.W.; Iervolino, E.; Yu, H.Y.; Wang, F.; Zhang, G.Q.; Sarro, P.M.; Zhang, G.Y. System in package (SiP) technology: Fundamentals, design and applications. *Microelectron. Int.* **2018**, *35*, 231–243. [CrossRef]
- Li, Y. Micro System Based on SiP Technology; Electronic Industry Press: Beijing, China, 2021; pp. 48–50.
- Appello, D.; Bernardi, P.; Grosso, M.; Reorda, M.S. System-in-package testing: Problems and solutions. *IEEE Des. Test Comput.* 2006, 23, 203–211. [CrossRef]
- 11. Chen, G.; Wang, G.; Hu, J.; Wang, K.; Xu, D. Survey on Chiplet Packaging Structure and Communication Structure. J. Comput. Res. Dev. 2022, 59, 22–30.
- 12. Lee, Q.Y.; Lee, M.X.; Lee, Y.C. A Hybrid Fuzzy Decision Model for Evaluating MEMS and IC Integration Technologies. *Micromachines* 2021, 12, 276. [CrossRef]
- 13. Xu, C.; Zhong, Z.W.; Choi, W.K. Evaluation of fan-out wafer level package strength. *Microelectron. Int.* 2019, 36, 54–61. [CrossRef]
- 14. Lau, J.H. Recent Advances and Trends in Advanced Packaging. *IEEE Trans. Compon. Packag. Manuf. Technol.* 2022, 12, 228–252. [CrossRef]

- 15. Yole Group-Follow the Latest Trend News in the Semiconductor Industry. Available online: https://www.yolegroup.com/ product/report/system-in-package-technology-and-market-trends-2021/ (accessed on 1 July 2021).
- Hou, S.Y.; Chen, W.C.; Hu, C.; Chiu, C.; Ting, K.C.; Lin, T.S.; Wei, W.H.; Chiou, W.C.; Lin, V.J.C.; Chang, V.C.Y.; et al. Wafer-Level Integration of an Advanced Logic-Memory System Through the Second-Generation CoWoS Technology. *IEEE Trans. Electron Devices* 2017, 64, 4071–4077. [CrossRef]
- 17. Stoppino, P.P.; Conci, A.; Lessio, T.; Ferrara, D. System in Package Feasibility Process. Proc. IEEE 2009, 97, 141–147. [CrossRef]
- Zhao, H.R.; Chen, M.X.; Peng, Y.; Wang, Q.; Kang, M.; Cao, L.H. TXV Technology: The cornerstone of 3D system-in-packaging. Sci. China-Technol. Sci. 2022, 65, 2031–2050. [CrossRef]
- Braun, T.; Becker, K.F.; Koch, M.; Bader, V.; Aschenbrenner, R.; Reichl, H.; IEEE. Flip chip molding-High reliable flip chip encapsulation. In Proceedings of the 52nd Electronic Components and Technology Conference (ECTC), San Diego, CA, USA, 28–31 May 2002; pp. 717–725.
- Liu, C.M.; Chen, L.S.; Romanowski, R.M. An electronic material flow control system for improving production efficiency in integrated-circuit assembly industry. *Int. J. Adv. Manuf. Technol.* 2009, 42, 348–362. [CrossRef]
- Zhang, X.S.; Wang, Q.; Wang, B.; Wang, G.; Gu, X.; Chen, Y.; Tan, L.; Zhou, S.J.; Cai, J. A Hybrid Antenna in Package Solution Using FOWLP Technology. J. Electron. Packag. 2022, 144, 6. [CrossRef]
- 22. Lau, J.H.; Li, M.; Qingqian, M.L.; Chen, T.; Xu, I.; Yong, Q.X.; Cheng, Z.; Fan, N.; Kuah, E.; Li, Z.; et al. Fan-Out Wafer-Level Packaging for Heterogeneous Integration. *IEEE Trans. Compon. Pack. Manuf. Technol.* **2018**, *8*, 1544–1560. [CrossRef]
- Chong, C.T.; Ho, S.W.; Choong, C.S.; Lim, P.S.S.; Bhattacharya, S. Heterogeneous Integration for Chiplets on FOWLP Development Line. In Proceedings of the 72nd IEEE Electronic Components and Technology Conference (ECTC), San Diego, CA, USA, 31 May–1 June 2022; pp. 655–659.
- Ali, M.; Hassan, A.; Honarparvar, M.; Nabavi, M.; Audet, Y.; Sawan, M.; Savaria, Y. A Versatile SoC/SiP Sensor Interface for Industrial Applications: Implementation Challenges. *IEEE Access* 2022, 10, 24540–24555. [CrossRef]
- Lu, B.; Zhu, H.R.; Duan, Z.M.; Dai, Y.F. A Wideband Millimeter Wave E-shape Antenna in Package with Embedded Wafer Level Ball Grid Array Technology. In Proceedings of the IEEE-MTT-S International Conference on Numerical Electromagnetic and Multiphysics Modeling and Optimization (NEMO), Hangzhou, China, 7–9 December 2020.
- Ahmed, F.; Furqan, M.; Stelzer, A. 120-GHz and 240-GHz Broadband Bow-Tie Antennas in eWLB Package for High Resolution Radar Applications. In Proceedings of the 48th European Microwave Conference (EuMC), Madrid, Spain, 25–27 September 2018; pp. 1109–1112.
- JCET-Fan-Out Wafer Level eWLB Technology as an Advanced System-in Package Solution. JCET Group-Technology. Available online: jcetglobal.com (accessed on 1 December 2019).
- JCET-Innovative Integration Solution for SiP Packages Using Fan-Out WLP. JCET Group-Technology. Available online: jcetglobal.com (accessed on 1 December 2019).
- Chiang, Y.P.; Tai, S.P.; Wu, W.C.; Yeh, J.; Wang, C.T.; Yu, D.C.H.; Soc, I.C. InFO\_oS (Integrated Fan-Out on Substrate) Technology for Advanced Chiplet Integration. In Proceedings of the 2021 IEEE 71st Electronic Components and Technology Conference (ECTC), San Diego, CA, USA, 1 June–4 July 2021; pp. 130–135.
- Tsai, M.H.; Hsu, S.K.; Shen, C.K.; Wei, P.S.; Chern, C.H.; Wu, T.L. A Miniature Electromagnetic Bandgap Structure Using Integrated Fan-Out Wafer-Level Package (InFO-WLP) for Gigahertz Noise Suppression. In Proceedings of the IEEE/MTT-S International Microwave Symposium, Philadelphia, PA, USA, 10–15 June 2018; pp. 1542–1544.
- 31. Amkor Technology, Inc. System in Package (SiP)-Amkor Technology; Amkor Technology, Inc.: Tampe, AZ, USA, 2021.
- Duan, G.; Kanaoka, Y.; McRee, R.; Nie, B.; Manepalli, R.; Soc, I.C. Die Embedding Challenges for EMIB Advanced Packaging Technology. In Proceedings of the 2021 IEEE 71st Electronic Components and Technology Conference (ECTC), San Diego, CA, USA, 1 June–4 July 2021; pp. 1–7.
- 33. Mahajan, R.; Sankman, R.; Patel, N.; Kim, D.W.; Aygun, K.; Qian, Z.G.; Mekonnen, Y.; Salama, I.; Sharan, S.; Iyengar, D.; et al. Embedded Multi-Die Interconnect Bridge (EMIB)-A High Density, High Bandwidth Packaging Interconnect. In Proceedings of the 66th IEEE Electronic Components and Technology Conference (ECTC), Las Vegas, NV, USA, 31 May–3 June 2016; pp. 557–565.
- Nawaz, A.A.; Khan, W.T.; Ulusoy, A.C. Organically Packaged Components and Modules Recent Advancements for Microwave and mm-Wave Applications. *IEEE Microw. Mag.* 2019, 20, 49–72. [CrossRef]
- 35. Tian, G.X.; Li, J.; Hou, F.Z.; Zhang, W.W.; Guo, X.P.; Cao, L.Q.; Wan, L.X. Design and Implementation of a Compact 3-D Stacked RF Front-End Module for Micro Base Station. *IEEE Trans. Compon. Pack. Manuf. Technol.* **2018**, *8*, 1967–1978. [CrossRef]
- Samanta, K.K. Cost-Effective Technologies for Next-Generation System on Package: Multilayer Transmission Lines and Interconnects for 5G and Millimeter-Wave. *IEEE Microw. Mag.* 2022, 23, 50–65. [CrossRef]
- Fang, Z.; Tang, G.; Lu, L.; Zeng, R.; Li, Z. Design of a Novel S-Band Bandpass Filter Based on Silicon Integrated Passive Device Technology. Semicond. Technol. 2018, 43, 347–352. [CrossRef]
- Su, Y.Q.; Yu, D.Q.; Ruan, W.B.; Jia, N. Development of Compact Millimeter-Wave Antenna by Stacking of Five Glass Wafers With Through Glass Vias. *IEEE Electron Device Lett.* 2022, 43, 934–937. [CrossRef]
- Yu, B.; Wang, Z.G.; Wu, P.; Li, O.P.; Cai, H.; He, J.; Wang, G.J.; Xu, R.M. High-Integration and Low-Cost Transmitter Packaging Solution for 0.2 THz SiP Application Using HTCC Technology. *IEEE Microw. Wirel. Compon. Lett.* 2022, 32, 680–683. [CrossRef]
- Bhutani, A.; Gottel, B.; Lipp, A.; Zwick, T. Packaging Solution Based on Low-Temperature Cofired Ceramic Technology for Frequencies Beyond 100 GHz. *IEEE Trans. Compon. Pack. Manuf. Technol.* 2019, 9, 945–954. [CrossRef]

- 41. Pendharkar, S. Smart Power Technologies Enabling Power SOC and SIP. In Proceedings of the 36th IEEE Symposium on VLSI Technology, Honolulu, HI, USA, 13–16 June 2016.
- 42. Lee, H.; Smet, V.; Tummala, R. A Review of SiC Power Module Packaging Technologies: Challenges, Advances, and Emerging Issues. *IEEE J. Emerg. Sel. Top. Power Electron.* 2020, *8*, 239–255. [CrossRef]
- 43. Hayashibe, S.; Murayama, K.; Bando, K.; Ito, H.; Suzuki, H.; Sugita, T. Evaluation of half-bridge power module with Pol-kW. In Proceedings of the 2021 IEEE CPMT Symposium Japan (ICSJ), Kyoto, Japan, 10–12 November 2021; pp. 130–133. [CrossRef]
- Tao, L.; WeiYin, W.; Jun, L. Random Vibration Response Analysis of Sip Power Supply Module. In Proceedings of the 2021 22nd International Conference on Electronic Packaging Technology (ICEPT), Xiamen, China, 14–17 September 2021; Volume 3, p. 3. [CrossRef]
- Tian, W.; Li, B.; Li, Z.; Cui, H.; Shi, J.; Wang, Y.; Zhao, J. Using Chiplet Encapsulation Technology to Achieve Processing-in-Memory Functions. *Micromachines* 2022, 13, 1790. [CrossRef]
- Liu, P.; Shi, S.; Xie, B.; Zhao, Z. Research on embedded integrate management unit for spacecraft avionics system. *Spacecr. Eng.* 2018, 27, 111–116. [CrossRef]
- Lin, M.S.; Huang, T.C.; Tsai, C.C.; Tam, K.H.; Hsieh, K.C.H.; Chen, C.F.; Huang, W.H.; Hu, C.W.; Chen, Y.C.; Goel, S.K.; et al. A 7-nm 4-GHz Arm(1)-Core-Based CoWoS(1) Chiplet Design for High-Performance Computing. *IEEE J. Solid State Circuit* 2020, 55, 956–966. [CrossRef]
- 48. Guo, X.; Xun, Q.; Li, Z.; Du, S. Silicon carbide converters and MEMS devices for high-temperature power electronics: A critical review. *Micromachines* **2019**, *10*, 406. [CrossRef]
- Wang, Z. Microsystems using three-dimensional integration and TSV technologies: Fundamentals and applications. *Microelectron*. Eng. 2019, 210, 35–64. [CrossRef]
- Jun, J.; Shin, S.; Kim, M.; Ahn, Y.; Kim, S. A +/- 0.15% RH Inaccuracy Humidity Sensing System With +/+/- +/- 0.44 degrees C (3 sigma) Inaccuracy On-Chip Temperature Sensor. *IEEE Sens. J.* 2021, 21, 2115–2123. [CrossRef]
- Robichaud, A.; Deslandes, D.; Cicek, P.V.; Nabki, F. A System in Package Based on a Piezoelectric Micromachined Ultrasonic Transducer Matrix for Ranging Applications. *Sensors* 2021, 21, 2590. [CrossRef] [PubMed]
- 52. Wang, L.H.; Fu, D.Y.; Zhao, J.Y.; Zhao, S.Q.; Wu, G.S.; Chen, H.Y.; Zhu, Y.F.; Yang, J.L. VCSEL Array Microsystem Packaging Based on SiP Technology. *Laser Optoelectron. Prog.* 2021, 58, 6. [CrossRef]
- Lu, M.J.; Mu, S.Y.; Cheng, C.S.; Chen, J. Advanced Packaging Technologies for Copackaged Optics. In Proceedings of the 2022 IEEE 72nd Electronic Components and Technology Conference (ECTC), San Diego, CA, USA, 31 May–3 June 2022; pp. 38–42. [CrossRef]
- Park, S.Y.; Nam, J.H.; Shim, J.N.; Lee, J.K.; Kwon, Y.T.; Lee, C.W.; Kim, J.H.; Kim, N.C. CiB(chip in board) optical engine module using advanced fan-out package technology. In Proceedings of the 2019 IEEE 69th Electronic Components and Technology Conference (ECTC), Las Vegas, NV, USA, 28–31 May 2019; pp. 563–568. [CrossRef]
- 55. Xue, J.; Razdan, S.; De Dobbelaere, P.; Miele, A.; Prasad, A.; Wang, T.; Patel, V.; Chadha, G. Trends and Challenges in Advanced Packaging Development for Silicon Photonics Beyond 400 Gbps in Hyperscale Data Center Networking Applications. In Proceedings of the 2022 6th IEEE Electron Devices Technology & Manufacturing Conference (EDTM), Oita, Japan, 6–9 March 2022; pp. 148–150. [CrossRef]
- 56. Ayotte, S.; Costin, F.; Babin, A.; Pare-Olivier, G.; Girard-Deschenes, E.; Morin, M.; Filion, B.; Bedard, K.; Chretien, P.; Bilodeau, G.; et al. Compact silicon photonics-based laser modules for FM-CW LIDAR and RFOG. *Proc. SPIE* **2020**, *11284*, 1128421–1128429. [CrossRef]
- Djogo, G.; Li, J.Z.; Ho, S.; Haque, M.; Ertorer, E.; Liu, J.; Song, X.L.; Suo, J.; Herman, P.R. Femtosecond laser additive and subtractive micro-processing: Enabling a high-channel-density silica interposer for multicore fibre to silicon-photonic packaging. *Int. J. Extrem. Manuf.* 2019, 1, 045002. [CrossRef]
- 58. Lee, C.C.; Lin, Y.M.; Liu, H.C.; Syu, J.Y.; Huang, Y.C.; Chang, T.C. Reliability evaluation of ultra thin 3D-IC package under the coupling load effects of the manufacturing process and temperature cycling test. *Microelectron. Eng.* **2021**, 244, 9. [CrossRef]
- Boriskov, P.; Ershova, N.; Putrolaynen, V.; Savitskii, M.; Seredov, P.; Soloviev, A.; Iop. Temperature simulation of system-inpackage produced with hybrid chip mounting technology. In Proceedings of the 3rd International Conference on Information Processing and Control Engineering (ICIPCE), Moscow, Russia, 4–7 August 2019.
- 60. Deng, S.S.; Hwang, S.J.; Lee, H.H. Temperature prediction for system in package assembly during the reflow soldering process. *Int. J. Heat Mass Transf.* **2016**, *98*, 1–9. [CrossRef]
- 61. Fu, G.C.; Su, Y.T.; Guo, W.D.; Wan, B.; Zhang, Z.Q.; Wang, Y. Life prediction methodology of system-in-package based on physics of failure. *Microelectron. Reliab.* **2018**, *88–90*, 173–178. [CrossRef]
- 62. Petrosyants, K.O.; Ryabov, N.I. Quasi-3D Thermal Simulation of Integrated Circuit Systems in Packages. *Energies* **2020**, *13*, 3054. [CrossRef]
- 63. Kwak, J.B.; Chung, S. Advanced utilization of 3D digital image correlation for thermal and impact reliabilities of electronics components. *Microelectron. Int.* 2021, 38, 14–22. [CrossRef]
- 64. Schmidt, C.; Altmann, F.; Naumann, F.; Lindner, A. Application of Lock-In-Thermography for 3d defect localisation in complex devices. In Proceedings of the 2nd Electronics System-Integration Technology Conference, Greenwich, UK, 1–4 September 2008; p. 1041.
- 65. Zhang, J.Y.; Zhou, B.; Li, G.Y. Study on Predicting the Temperature of Stacked Chip based on Thermal Resistance Matrix. In Proceedings of the 24th IEEE International Symposium on the Physical and Failure Analysis of Integrated Circuits (IPFA), Chengdu, China, 4–7 July 2017.

- 66. Yuile, A.; Schulz, A.; Muller, J.; Wiese, S. Analysis of selective bonding processes using reactive multi-layers for system integration on LTCC based SiPs. *Microsyst. Technol.* **2022**, *28*, 1995–2009. [CrossRef]
- 67. Shih, M.K.; Lai, W.H.; Liao, T.; Chen, K.; Chen, D.L.; Hung, C.P. Thermal and Mechanical Characterization of 2.5-D and Fan-Out Chip on Substrate Chip-First and Chip-Last Packages. *IEEE Trans. Compon. Pack. Manuf. Technol.* 2022, 12, 297–305. [CrossRef]
- 68. Bocca, A.; Macii, A. Thermal modeling and analysis of a power ball grid array in system-in-package technology. *Multiscale Multidiscip. Model Exp. Des.* **2022**, *5*, 31–41. [CrossRef]
- 69. Jiang, T.F.; Ryu, S.K.; Zhao, Q.; Im, J.; Huang, R.; Ho, P.S. Measurement and analysis of thermal stresses in 3D integrated structures containing through-silicon-vias. *Microelectron. Reliab.* **2013**, *53*, 53–62. [CrossRef]
- Wu, C.L.; Huang, R.; Liechti, K.M. Characterizing Interfacial Sliding of Through-Silicon-Via by Nano-Indentation. *IEEE Trans. Device Mater. Reliab.* 2017, 17, 355–363. [CrossRef]
- Feng, X.; Cao, H.; Yu, H.; Gao, L.; Li, M. Study of international on electroplating copper used in through silicon viafilling. In Proceedings of the 2011 International Conference on Electronic Packaging Technology and High Density Packaging, Shanghai, China, 8–11 August 2011; pp. 1018–1021.
- Winiarski, B.; Withers, P.J. Micron-Scale Residual Stress Measurement by Micro-Hole Drilling and Digital Image Correlation. *Exp. Mech.* 2012, 52, 417–428. [CrossRef]
- 73. Yu, S.Y.; Kwon, Y.M.; Kim, J.; Jeong, T.; Choi, S.; Paik, K.W. Studies on the Thermal Cycling Reliability of BGA System-in-Package (SiP) with an Embedded Die. *IEEE Trans. Compon. Packag. Manuf. Technol.* **2012**, *2*, 625–633. [CrossRef]
- 74. Sa, Z.; Wang, S.; Feng, J.; Ma, J.; Zhang, N.; Yu, G.; Liang, J.; Tian, Y. Simulation of SiP Solder Joint Geometry and Random Vibration Reliability Prediction. *J. Mech. Eng.* **2022**, *58*, 276–283.
- 75. Fruehauf, P.; Munding, A.; Pressel, K.; Schwarz, P.; Vogt, M. Chip-Package-Board Reliability of System-in-Package Using Laminate Chip Embedding Technology Based on Cu Leadframe. *IEEE Trans. Compon. Packag. Manuf. Technol.* 2020, 10, 44–56. [CrossRef]
- 76. Vinciguerra, V.; Malgioglio, G.L.; Landi, A. Modelling the Elastic Energy of a Bifurcated Wafer: A Benchmark of the Analytical Solution vs. The ANSYS Finite Element Analysis. *Compos. Struct.* **2022**, *281*, 14996. [CrossRef]
- 77. van Dijk, M.; Huber, S.; Stegmaier, A.; Walter, H.; Wittler, O.; Schneider-Ramelow, M. Experimental and simulative study of warpage behavior for fan-out wafer-level packaging. *Microelectron. Reliab.* **2022**, *135*, 9. [CrossRef]
- Moore, L.; Barrett, J. Board-Folding Method for Fabrication of 3-D System in Package Devices. *IEEE Trans. Compon. Packag. Manuf. Technol.* 2012, 2, 1209–1216. [CrossRef]
- Lim, H.; Fuentes, R. Efficient Decoupling Capacitor Optimization for Subsystem Module Package. J. Microelectron. Packag. Soc. 2022, 29, 114585.
- 80. Li, Z.; Lyu, L.; Huang, X.; Wang, P.; Zhong, W. Progressive Optimization Method for RF SiP System. J. Microw. 2018, 34, 29–35.
- Kang, J.; Chae, K.; Jeong, J. Connectivity Test for System in Package Interconnects. J. Electron. Test.-Theory Appl. 2019, 35, 559–565. [CrossRef]
- Yang, L.; Yu, Z.; Wei, J. Design Technique of a SIP Test and Debugging System Based on JTAG Interface. *Semicond. Technol.* 2018, 43, 316–320. [CrossRef]
- Li, Y.; Li, J.L.; Guo, Y.X.; He, C.H.; Li, R.B.; Chen, W.; Liao, W.L.; Li, Y.H.; Li, P.; Liu, J.X.; et al. Experimental Study of Transient Dose Rate Effect on System-in-Package SZ0501. *IEEE Trans. Nucl. Sci.* 2022, 69, 1840–1849. [CrossRef]
- Hu, S.; Xu, X.; Zheng, Y.; Zhao, Z.; Wang, W.; Xu, R.; An, H.; Mo, Z. Algorithms in Linear Solver for Large-scale Time-harmonic Maxwell's Equations in SiP Applications: State-of-the-art and Challenges. *Chin. J. Comput. Phys.* 2021, 38, 131–145.
- 85. Ding, L.; Wei, X.C.; Tang, Z.Y.; Wen, J.; Gao, L.; Gao, R.X.K. Near-Field Scanning Based Shielding Effectiveness Analysis of System in Package. *IEEE Trans. Compon. Pack. Manuf. Technol.* 2021, *11*, 1235–1242. [CrossRef]
- 86. Houzet, G.; Lacrevaz, T.; Bermond, C.; Artillan, P.; Auchere, D.; Schwarz, L.; Flechet, B. In-situ characterization up to 100 GHz of insulators used in new 3D "System in Package on board" (SiPoB) technologies. *Microelectron. J.* 2021, 108, 7. [CrossRef]
- 87. Chen, Y.; Zhang, P.; Xia, K. Process Defect Diagnosis and Analysis of the TSV. Semicond. Technol. 2018, 43, 473–479. [CrossRef]
- 88. Li, Z.; Miao, M. Study and Performance Analysis of Bar TSV Structure. J. Syst. Simul. 2018, 30, 2044–2049. [CrossRef]
- Wang, M.C.; Ma, S.L.; Jin, Y.F.; Wang, W.; Chen, J.; Hu, L.L.; He, S.W. A RF Redundant TSV Interconnection for High Resistance Si Interposer. *Micromachines* 2021, 12, 169. [CrossRef] [PubMed]
- Burggraf, J.; Uhrmann, T.; Pires, M. Collective Die Bonding: An Enabling Toolkit for Heterogeneous Integration. ECS Trans. 2020, 98, 173–180. [CrossRef]
- 91. Rajan, S.K.; Kaul, A.; Sarvey, T.E.; May, G.S.; Bakir, M.S. Monolithic Microfluidic Cooling of a Heterogeneous 2.5-D FPGA With Low-Profile 3-D Printed Manifolds. *IEEE Trans. Compon. Pack. Manuf. Technol.* **2021**, *11*, 974–982. [CrossRef]
- Guo, W.; Chen, C.; Yang, Y.; Xie, J. Design of Micro-Channel Layout on 3D System in Package Based on Fractal Theory. Semicond. Technol. 2018, 43, 364–368. [CrossRef]
- 93. Zhang, J.W.; Fan, Y.L.; Cheng, J.J.; Wu, L.; Xu, Y.J. A Programmable Pulse Sequence Generator for High Temperature Low-Field NMR Apparatus. *Instrum. Exp. Technol.* **2018**, *61*, 506–515. [CrossRef]
- Chieh, K. Robust Design Analysis on Fatigue Life of Lead-free Sn0.5Ag Solder in a Multichip Module Package. *Appl. Mech. Mater.* 2013, 284–287, 375–379. [CrossRef]
- 95. Chen, C.; Hou, F.Z.; Liu, F.M.; She, Q.; Cao, L.Q.; Wan, L.X. Thermo-mechanical reliability analysis of a RF SiP module based on LTCC substrate. *Microelectron. Reliab.* 2017, *79*, 38–47. [CrossRef]

- 96. Tang, Y.; Luo, S.M.; Li, G.Y.; Yang, Z.; Chen, R.; Han, Y.; Hou, C.J. Optimization of the thermal reliability of a four-tier die-stacked SiP structure using finite element analysis and the Taguchi method. *Microelectron. J.* **2018**, *73*, 18–23. [CrossRef]
- 97. Rangarajan, S.; Hadad, Y.; Choobineh, L.; Sammakia, B. Minimizing Temperature Nonuniformity by Optimal Arrangement of Hotspots in Vertically Stacked Three-Dimensional Integrated Circuits. J. Electron. Packag. 2020, 142, 9. [CrossRef]
- Wu, P.; Liu, F.; Li, J.; Chen, C.; Hou, F.; Cao, L.; Wan, L. Design and implementation of a rigid-flex RF front-end system-in-package. Microsyst. Technol. 2017, 23, 4579–4589. [CrossRef]
- Lau, J.H.; Yue, T.G. Effects of TSVs (through-silicon vias) on thermal performances of 3D IC integration system-in-package (SiP). Microelectron. Reliab. 2012, 52, 2660–2669. [CrossRef]
- Zhong, Y.; Sun, G.; Wang, N.; Fang, J.; Cui, S. Study on pin brazing strength of AIN package. *Electron. Compon. Mater.* 2019, 38, 63–67, 73. [CrossRef]
- Kim, Y.; Park, S.M.; Kim, S.E. Effect of Ag Nanolayer in Low Temperature Cu/Ag-Ag/Cu Bonding. J. Microelectron. Packag. Soc. 2021, 28, 59–64.
- Seo, H.; Sung, P.H.; Kim, G.; Park, Y.-B.; Eunkyung, K.S. Bonding Strength Evaluation of Copper Bonding Using Copper Nitride Layer. J. Microelectron. Packag. Soc. 2020, 27, 55–60.
- Seo, H.; Park, H.; Kim, S.E. Comprehensive Analysis of a Cu Nitride Passivated Surface That Enhances Cu-to-Cu Bonding. *IEEE Trans. Compon. Packag. Manuf. Technol.* 2020, 10, 1814–1820. [CrossRef]
- Zhou, J.; Zhang, X.; Zhong, L. 3D Integration Design of Millimeter Wave Module Based on BGA Interconnection. *Telecommun. Eng.* 2019, 59, 724–728. [CrossRef]
- Craton, M.T.; Konstantinou, X.; Albrecht, J.D.; Chahal, P.; Papapolymerou, J. Additive Manufacturing of a W-Band System-on-Package. *IEEE Trans. Microw. Theory Technol.* 2021, 69, 4191–4198. [CrossRef]
- Tehrani, B.K.; Tentzeris, M.M. Fully Inkjet-Printed Ramp Interconnects for Wireless Ka-Band MMIC Devices and Multi-Chip Module Packaging. In Proceedings of the 48th European Microwave Conference (EuMC), Madrid, Spain, 25–27 September 2018; pp. 1037–1040.
- Bellaredj, M.L.F.; Pardue, C.A.; Kohl, P.; Swaminathan, M. Fabrication of package embedded spiral inductors with two magnetic layers for flexible SIP point of load converters in Internet of Everything devices. *Microelectron. Eng.* 2018, 189, 18–27. [CrossRef]
- Bellaredj, M.L.F.; Davis, A.K.; Kohl, P.; Swaminathan, M.; Sandler, S. Design, Fabrication, and Characterization of Package Embedded Solenoidal Magnetic Core Inductors for High-Efficiency System-In-Package Integrated Voltage Regulators. *IEEE Trans. Magn.* 2019, 55, 7. [CrossRef]
- Zhi, C.L.; Dong, G.; Zhu, Z.M.; Yang, Y.T. A TSV-Based 3-D Electromagnetic Bandgap Structure on an Interposer for Noise Suppression. *IEEE Trans. Compon. Packag. Manuf. Technol.* 2022, 12, 147–154. [CrossRef]
- Xu, K.; Vaisband, B.; Sizikov, G.; Li, X.; Friedman, E.G. EMI Suppression with Distributed LLC Resonant Converter for High-Voltage VR-on-Package. *IEEE Trans. Compon. Packag. Manuf. Technol.* 2020, 10, 263–271. [CrossRef]
- Xu, K.; Vaisband, B.; Sizikov, G.; Li, X.; Friedman, E.G. Power Noise and Near-Field EMI of High-Current System-in-Package with VR Top and Bottom Placements. *IEEE Trans. Compon. Packag. Manuf. Technol.* 2019, 9, 712–718. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.